Author's response to reviews

Title: Physical activity for people with dementia: a scoping study

Authors:

Alison Bowes (a.m.bowes@stir.ac.uk)
Alison Dawson (a.s.f.dawson@stir.ac.uk)
Ruth Jepson (ruth.jepson@ed.ac.uk)
Louise McCabe (louise.mccabe@stir.ac.uk)

Version: 2  Date: 27 October 2013

Author's response to reviews:

Changes were requested by Referee 2 and the editor.
Referee 1 did not request changes.

In response to Referee 2, we have made the following specific changes:

1. Page 6, 2nd paragraph, line 3: The acronyms have been spelled out.
2. Page 6, 3rd paragraph: The abbreviations have been placed in brackets, and brackets removed from the full names.
3. Page 7, 1st paragraph, line 6: ‘clarified’ has been changed to ‘described’.
4. Table 1: numbers in this table, including those for Controlled Before and After design, have been checked and corrected.
5. Table 8, the numbers for gender have been corrected.
6. Page 13, 2nd paragraph, line 1: AHPs has been spelled out.
7. Page 14, 1st paragraph, line 8: 42% has been spelled out as it is at the beginning of the sentence.

We have also re-read the paper and improved the language where possible.

We have also made the following formatting changes as requested:

1. We have included a title page at the front of the manuscript.
2. We have removed the duplicate figures and tables.

We have checked the manuscript against the journal style, and believe that it conforms to requirements. We used the template from the journal website and followed the authors’ checklist for manuscript formatting.

Yours sincerely

Alison Bowes